Author's response to reviews

Title: Proliferation and survival molecules implicated in the inhibition of BRAF pathway in thyroid cancer cells harbouring different genetic mutations.

Authors:

Ana Preto (apreto@ipatimup.pt)
Joana Gonçalves (jgoncalves@cnc.cj.uc.pt)
Ana P Rebocho (arebocho@ipatimup.pt)
Joana Figueiredo (jfigueiredo@ipatimup.pt)
Ana M Meireles (ameireles@ipatimup.pt)
Ana S Rocha (arocha@ipatimup.pt)
Helena M Vasconcelos (hvasconcelos@ipatimup.pt)
Hugo Seca (hteixeira@ipatimup.pt)
Raquel Seruca (rseruca@ipatimup.pt)
Paula Soares (psoares@ipatimup.pt)
Manuel Sobrinho-Simões (ssimoes@ipatimup.pt)

Version: 6 Date: 27 October 2009

Author's response to reviews: see over
To:

Professor Melissa Norton  
Editor-in-Chief of BMC Cancer

Dear Professor Melissa Norton,

Please find attached electronic version of the final version of the manuscript entitled: “Proliferation and survival molecules implicated in the inhibition of BRAF pathway in thyroid cancer cells harbouring different genetic mutations”.

Following the instructions of the BioMed Central Editorial Production Team, we corrected the list of authors in the manuscript in order to be written exactly as they are in the submission system. We also completed the reference 16 and 33 and removed the additional files from the system. The manuscript text was carefully read and small final corrections were made, the text is now presented in black and the image files do not include figure number.

Sincerely yours,

Paula Soares, MD, PhD  
IPATIMUP, Rua Dr. Roberto Frias, s/n- 4200-465 Porto, Portugal  
Phone: +351.225570700, Fax: +351.225570799, e-mail:psoares@ipatimup.pt